Author’s response to reviews

Title: A case of posterior reversible encephalopathy syndrome secondary to asymptomatic post streptococcal glomerulonephritis in a child with sickle cell anemia: a case report.

Authors:

ehab Hanafy (ehab.hmahmoud@gmail.com)
Duaa Alshareef (wrdha2012@hotmail.com)
Suhaila Osman (suhailasalah@hotmail.com)
Abdullah Al Jabri (Dr.aaljabri@gmail.com)
Faisal Nazim (nazim_prof@yahoo.com)
Gihan Mahmoud (drgihanmahmoud@gmail.com)

Version: 3 Date: 26 Dec 2017

Author’s response to reviews:

Thank you for your comments.

All changes are highlighted within the new submitted text by yellow color

1- A paragraph is added at the end of introduction stating the rarity of the case and message behind presentation

2- The case presentation is formulated in the required fashion

3- Detailed vital signs are included

4- Phrase “no history of smoking nor alcohol consumption was deleted”

5- Results of urine analysis were included as new table [table 2]

6- Last paragraph in case presentation section included the follow up and results of recent labs and MRI brain which were normal.

7- Abbreviations section is added